Author’s response to reviews

Title: Antioxidant Vitamins A, C, E and Coenzyme Q10 vs Dexamethasone: Comparisons of Their Effects in Pulmonary Contusion Model.

Authors:

Mertol Gokce (dr.mertol@gmail.com)
Ozkan Saydam (ozkan.saydam@gmail.com)
Volkan Hanci (vhanci@gmail.com)
Murat Can (drcanmurat@yahoo.com)
Burak Bahadir (burakbahadir@yahoo.com)

Version: 2  Date: 25 July 2012

Author’s response to reviews: see over
Dear Editorial Team Member, Vipin Zamvar,

We have addressed the requested changes in the reviewer’s suggestions. However all of the requested changes were related to style of the manuscript, i.e. writing style of the references, dots and commas etc. In the revised form we upload a manuscript with the changes highlighted in red and a clear version of the same. Thank you for your assistance.

Best Regards,

Mertol Gokce